Author's response to reviews

Title: Strategies for obtaining unpublished drug trial data: A qualitative interview study

Authors:

Nicole E Wolfe (nicwolfe@gmail.com)
Peter C Gøtzsche (pcg@cochrane.dk)
Lisa Bero (berol@pharmacy.ucsf.edu)

Version: 3 Date: 22 April 2013

Author's response to reviews:

Response to editor:

"The authors should respond to editorial comments but still adhere to the abstract's word limit. The abstract should be revised by giving a couple of examples or a little more detail while ensuring that all of the additional information is included in the main text of the manuscript.

The origin of Table 2's content is now well described in the text but it would be helpful to reflect the source of information in the title of the table or in the individual table headings."

RESPONSE: We have edited the abstract, which is now 349 words and adheres to the 350 word limit.
We have edited the title of Table 2 as follows: Table 2: Advantages and disadvantages of sources of unpublished data described by interview respondents

Thank you for your consideration of the manuscript.

Sincerely,
Lisa Bero